NuMI: 9AM Meeting

- Tue Nov 29th: Ramp card replaced for QD103: elog<u>228749</u>, elog<u>228762</u>, elog<u>228778</u> & elog<u>228850</u>
 - Thanks to controls
- Tue Nov 29th: TSD took advantage of down day to do maintenance work
- Tue Nov 29th: Replaced digitizer for a button BPM
- Wed Nov 30th: TSD made an access (beam off) to resolve differential pressure issue: elog228935 & elog228958
 - Repaired a cracked hose & replaced filters
 - Thank you to TSD for staying late and finishing the work
- Wed Nov 30th: Replaced crate for a button BPM
 - Thanks to instrumentation
- Up Next: Implementing a long-term solution for updating permit
 - Try tightening current windows on all correctors
 - Thanks to EE Support for working with us on developing a solution

